Author's response to reviews

Title: Factors influencing the implementation of a mental health guideline: an exploratory investigation using psychological theory

Authors:

Susan Michie (s.michie@ucl.ac.uk)
Stephen Pilling (s.pilling@ucl.ac.uk)
Philippa Garety (p.garety@iop.kcl.ac.uk)
Paula Whitty (p.m.whitty@ncl.ac.uk)
Martin P Eccles (martin.eccles@ncl.ac.uk)
Marie Johnston (m.johnston@abdn.ac.uk)
Jemma Simmons (jemmareiff@hotmail.com)

Version: 3 Date: 21 December 2006

Author's response to reviews:

Dear Brian,

I have made the requested formatting changes and also those asked by Reviewer 2. Duplicate mention of coding reliability was removed from the Results section; the reference to Table 2 in relation to kappa was removed, thus correcting the order of Tables mentioned in the text. However, I was unable to find the repeated text on page 9 that the reviewer refers to.

With best wishes,

Susan